Author's response to reviews

Title: Effectiveness of the bucco-lingual technique within a school-based supervised toothbrushing program on preventing caries: a randomized controlled trial

Authors:

Paulo Frazão (pafrazao@usp.br)

Version: 5 Date: 9 February 2011

Author's response to reviews: see over
Dear editor:

I am writing to present the final version of the original article entitled *Effectiveness of the bucco-lingual technique within a school-based supervised toothbrushing program on preventing caries: a randomized controlled trial*. I added an Authors' contributions section.

The research was supported by The Foundation for the Support of Research of the State of São Paulo, Brazil (Grant 51300-0/2006) and the Department of Health from Sao Vicente City Hall. There was not any financial concern or other relationships that might lead to a conflict of interest.

The manuscript was reviewed, has not been published before, and is not being considered for publication elsewhere. I agree to assign exclusive copyright to *BMC Oral Health*, if and when the manuscript is accepted for publication. All changes were indicated by letters in boldface.

Yours faithfully,

![Signature]

Paulo Frazão

Correspondence to Professor Paulo Frazão, Faculdade de Saúde Pública da Universidade de São Paulo. Av. Dr Arnaldo 715 – São Paulo, SP, Brasil. CEP 01246-905 Tel: (55 11) 3061-7957 Fax: (55 11) 3083-3501 Email: pafrazao@usp.br